Author's response to reviews

Title: Do self-rated oral health and dental care utilisation depend on socioeconomic factors? Recent evidence from a large population-based study in Sweden

Authors:

Anu Molarius (anu.molarius@ltv.se)
Sevek Engström (sevek.engstrom@ltv.se)
Håkan Flink (hakan.flink@ltv.se)
Bo Simonsson (bo.simonsson@ltv.se)
Åke Tegelberg (ake.tegelberg@ltv.se)

Version: 3 Date: 27 May 2014

Author's response to reviews:

The line numbers are now continuous.